Author's response to reviews

Title: Indoor solid fuel use and tuberculosis in China: a matched case-control study

Authors:

Xiaohong Kan (ahfidelis@126.com)
Chen-Yuan Chiang (cychiang@theunion.org)
Donald A Enarson (denarson@theunion.org)
Wenhua Chen (hyxcwh@163.com)
Jianan Yang (43567@ah163.com)
Genwang Chen (chengenwang@126.com)

Version: 4 Date: 23 June 2011

Author's response to reviews:

We thank Editors’ comments. We provide point-by-point response to Editors’ comments.

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

Answer: we have done so.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Answer: we have removed all highlighting and tracking from the manuscript.

Author list: Please place all the superscript in the same font

Answer: it is in the same font.

Headers: Please remove "Indoor air pollution and tuberculosis"

Answer: It is removed.

References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

Answer: We have made revision on citation.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Answer: We have done so.

Once again, we thank the editors for their comments on this manuscript.